Author’s response to reviews

Title: Testing the feasibility of a mobile technology intervention promoting healthy gestational weight gain in pregnant women (txt4two): a study protocol for a randomised controlled trial

Authors:

Jane C Willcox (jwillcox@deakin.edu.au)
Karen J Campbell (kcamp@deakin.edu.au)
Elizabeth A McCarthy (eamcca@unimelb.edu.au)
Shelley A Wilkinson (shelley.wilkinson@mater.org.au)
Martha Lappas (mlappas@unimelb.edu.au)
Kylie Ball (kylie.ball@deakin.edu.au)
Briana Fjeldsoe (b.fjeldsoe@sph.uq.edu.au)
Anne Griffiths (anne.griffiths@deakin.edu.au)
Robyn Whittaker (r.whittaker@nihi.auckland.ac.nz)
Ralph Maddison (r.maddison@nihi.auckland.ac.nz)
Alexis Shub (ashub@internode.on.net)
Deborah Pidd (DPidd@mercy.com.au)
Elise Fraser (EFRaser@mercy.com.au)
Nelly Moshanos (NMoshonas@mercy.com.au)
David A Crawford (david.crawford@deakin.edu.au)

Version: 3
Date: 15 April 2015

Author’s response to reviews:

Review letter attached "Review letter to the editor_Trials 2015(2).docx"